Author's response to reviews

Title: Participation in everyday activities and quality of life in pre-teenage children living with cerebral palsy in South West Ireland.

Authors:

Vicki M McManus (v.mcmanus@ucc.ie)
Paul Corcoran (paul.nsr@iol.ie)
Ivan J Perry (I.Perry@ucc.ie)

Version: 5 Date: 30 September 2008

Author's response to reviews:

Dear Editor,

Many thanks for accepting our paper subject to revision of minor elements. We have corrected the typos and the comment from the reviewer on changing verbage on GMFCS.

the paper is now in its final edit and we thank you for your efforts at getting the paper to this standard.

Kind regards,

Vicki McManus (On behalf of the team).